

# Impact of the 2021 European Society for Cardiology prevention guideline's stepwise approach for cardiovascular risk factor treatment in patients with established atherosclerotic cardiovascular disease

Joris Holtrop (1), Deepak L. Bhatt (1)<sup>2</sup>, Kausik K. Ray<sup>3</sup>, François Mach<sup>4</sup>, Yvo M. Smulders<sup>5</sup>, David Carballo (1)<sup>4</sup>, Philippe Gabriel Steg<sup>6</sup>, Frank L.J. Visseren<sup>1</sup>, and Jannick A.N. Dorresteijn (1)<sup>1</sup>\*; on behalf of the SMART study group

<sup>1</sup>Department of Vascular Medicine, University Medical Centre Utrecht, Heidelberglaan 100, Utrecht, 3584 CX, The Netherlands; <sup>2</sup>Mount Sinai Fuster Heart Hospital, Icahn School of Medicine at Mount Sinai Health System, New York, NY, USA; <sup>3</sup>Imperial Centre for Cardiovascular Disease Prevention, ICTU-Global, Imperial College London, London, UK; <sup>4</sup>Division of Cardiology, Faculty of Medicine, Geneva University Hospitals, Geneva, Switzerland; <sup>5</sup>Department of Internal Medicine, VU University Medical Center, Amsterdam, The Netherlands; and <sup>6</sup>Department of Cardiology, Université Paris-Cité, FACT (French Alliance for Cardiovascular Trials) NSERM1148/LVTS, AP-HP, Hôpital Bichat, Paris, France

Received 16 November 2023; revised 24 January 2024; accepted 27 January 2024; online publish-ahead-of-print 7 February 2024

Aims	This study aimed to evaluate the stepwise approach for cardiovascular (CV) risk factor treatment as outlined by the European Society for Cardiology 2021 guidelines on CV disease (CVD) prevention in patients with established atheroscler- otic CVD (ASCVD).
Methods and results	In patients with ASCVD, included in UCC-SMART ( $n = 8730$ ) and European parts of the REACH registry ( $n = 18$ 364), the 10-year CV risk was estimated using SMART2. Treatment effects were derived from meta-analyses and trials. Step 1 recommendations were LDL cholesterol (LDLc) < 1.8 mmol/L, systolic blood pressure (SBP) < 140 mmHg, using any antithrombotic medication, sodium–glucose co-transporter 2 (SGLT2) inhibition, and smoking cessation. Step 2 recommendations were LDLc < 1.4 mmol/L, SBP < 130 mmHg, dual-pathway inhibition (DPI, aspirin plus low-dose rivaroxaban), colchicine, glucagon-like peptide (GLP)-1 receptor agonists, and eicosapentaenoic acid. Step 2 was modelled accounting for Step 1 non-attainment. With current treatment, residual CV risk was 22%, 32%, and 60% in the low, moderate, and pooled (very) high European risk regions, respectively. Step 2 could prevent up to 198, 223 and 245 events per 1000 patients treated, respectively. Following Step 2, the number of patients with a CV risk of <10% could increase from 20%, 6.4%, and 0.5%, following Step 1, to 63%, 48%, and 12%, in the respective risk regions.
Conclusion	With current treatment, residual CV risk in patients with ASCVD remains high across all European risk regions. The inten- sified Step 2 treatment options result in marked further reduction of residual CV risk in patients with established ASCVD.
Lay summary	Patients with established cardiovascular disease are at high risk for new cardiovascular events. The European Society of Cardiology guideline for the prevention of cardiovascular disease introduced a stepwise treatment approach. Step 1 in this approach are treatments that apply to all patients, and Step 2 are intensive treatments that can be prescribed to patients who are still at high risk of new events even with Step 1 treatments. The current study investigates the effect of Steps 1 and 2 on the risk of cardiovascular disease in 27 094 patients all across Europe. With the conventional treatments of Step 1 the risk

\* Corresponding author. Tel: +31 88 75 555 55, Fax: +31 88 75 555 14, Email: J.A.N.Dorresteijn-2@umcutrecht.nl

© The Author(s) 2024. Published by Oxford University Press on behalf of the European Society of Cardiology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

of cardiovascular disease remains high, with a 10-year risk of new events higher than 10% in 80–99% of patients. The intensive treatment options from Step 2 could prevent additional 198–245 new cardiovascular events for every 1000 patients that are treated. With intensive treatment, up to 63% of patients could achieve a 10-year risk of new cardiovascular disease below 10%.

### **Key findings**

Guideline-recommended intensive treatment of patients with cardiovascular disease could prevent additional 198–245 new cardiovascular events for every 1000 patients treated.

#### **Graphical Abstract**



## Introduction

Patients with established atherosclerotic cardiovascular (CV) disease (ASCVD) are at high risk for CV events.<sup>1</sup> This residual risk is highly variable among patients with ASCVD and is dependent on the degree of attainment of treatment goals.<sup>1</sup> While this risk of recurrence can be reduced by treatment of modifiable risk factors, a substantial number of patients remain at high residual risk of CV events, even with perfect attainment of guideline-recommended treatment goals.<sup>1-3</sup> In the European Society for Cardiology (ESC) guideline on cardiovascular disease (CVD) prevention in clinical practice, the well-established strategies and recently explored novel treatments aimed at reducing this residual risk have been combined into a two-step approach for risk factor treatment.<sup>4</sup> In this approach, the Step 1 treatments are the general preventive measures that apply to all patients with ASCVD, including lifestyle recommendations and target values such as a LDL cholesterol (LDLc) < 1.8 mmol/L and systolic blood pressure (SBP) between 130 and 140 mmHg. Additionally, the guideline recommends considering Step 2 treatments, also referred to as intensified treatment options, on an individual basis using predicted residual 10-year or lifetime CV risk, and taking into account comorbidities, frailty, and patient preferences. Step 2 includes stricter control of certain targets such as LDLc reduction to <1.4 mmol/L and lowering of SBP to 120-<130 mmHg, as well as novel interventions such as dual-pathway inhibition (DPI).

While the Step 2 interventions have been proven to reduce the risk of CV events in clinical trials, the real-world impact of their combined implementation according to guideline recommendations has not yet been investigated.

The aim of the current study is to provide insight into the impact of the 2021 ESC guideline's stepwise approach to CVD prevention in clinical practice on residual 10-year risk of CV events in patients with ASCVD in each of the European risk regions.

# Methods

Participants from the Utrecht Cardiovascular Cohort-Secondary Manifestations of Arterial Disease (UCC-SMART), included between September 1996 and January 2020, and from the European countries within the Reduction of Atherothrombosis for Continued Health (REACH) registry, included between December 2003 and June 2004, with established ASCVD at baseline were eligible for analyses. Descriptions of both are published elsewhere.<sup>5,6</sup> Briefly, all measurements are standardized (e.g. fasting lipid measurements) and assessed at baseline. Atherosclerotic CVD at baseline was defined as coronary artery disease (CAD), cerebrovascular disease (PAD) as per study definition (see Supplementary material online, *Table S1*).

#### **Risk estimation**

In accordance with guideline recommendations, the SMART2 risk score with region-specific recalibration factors was used to predict 10-year risk

of CV events.<sup>4,7</sup> In line with the intended use of SMART2, the analyses were restricted to patients aged between 40 and 80 years old. Risk regions are defined as per SMART2-definitions.<sup>7</sup> Due to their limited population sizes individually, the high- and very high-risk regions were pooled. Since LDLc, HDL cholesterol (HDLc), C-reactive protein, and years since first CV event were unavailable in the REACH registry, these data were imputed using multilevel single imputation with predictive mean matching based on pooled REACH and SMART data. Convergence of the imputation model was assessed visually.

#### **Treatment effects**

Treatment targets were derived from the 2021 ESC guideline on CVD prevention in clinical practice.<sup>4</sup> Supplementary material online, Table S2 lists an overview of the assumptions made in modelling treatment effects.<sup>8–14</sup> For analyses purposes, treatment targets for SBP were translated to reaching the midpoint of the target range, i.e. 135 mmHg in Step 1 and 125 mmHg in Step 2, unless the SBP was already lower. Step 1 LDLc target was translated to reduction to 1.7 mmol/L, unless the LDLc was already lower. Since information on intensity of lipid-lowering therapy was unavailable, Step 2 LDLc reduction was modelled as an additional 30% decrease in LDLc, for those patients with an LDLc of  $\geq$ 1.4 mmol/L. This reduction might be achievable through (a combination of) higher intensity statins, addition of ezetimibe and bempedoic acid, or proprotein convertase subtilisin/kexin type 9 (PCSK9) inhibition, which could feasibly result in a 59% additional LDLc reduction.  $^{15-18}$  For patients with diabetes mellitus type 2 (DM) with ASCVD, Step 1 was modelled as initiation of sodium-glucose co-transporter-2 inhibitors (SGLT2i). Step 1 lifestyle effects other than smoking cessation were not included in analyses since unambiguous targets and efficacy measures are lacking, and effects are likely to be mediated by changes in other risk factors (e.g. blood pressure).<sup>19,20</sup> An exploratory analysis was performed to investigate the potential impact of increasing physical activity and adhering to the Mediterranean diet as part of extended lifestyle interventions in Step 1 (see Supplementary methods). No explicit preference is expressed in the guideline with regard to intensive antithrombotic treatment in Step 2. For analyses purposes, this was modelled as DPI (i.e. adding low-dose rivaroxaban to aspirin) which has been evaluated in patients with CAD and PAD and therefore applies to more patients than dual antiplatelet therapy (DAPT). Correspondingly, the effect of colchicine was only modelled for patients with a history of CAD and the effect of eicosapentaenoic acid (EPA) was only modelled for patients with fasting triglycerides levels  $\geq$ 150 and <500 mg/ dL (1.52–5.63 mmol/L). For patients with DM and ASCVD, Step 2 was modelled as initiation of glucagon-like peptide 1 receptor agonists (GLP1-RA).

### **Scenarios**

Two scenarios for Step 2 implementation were investigated. The first assumed imperfect attainment of Step 1 targets, illustrating the possible realworld impact of Step 2. The second scenario assumed perfect attainment of all Step 1 targets which, while unrealistic in the real-world, demonstrates the most cautious estimate for the effect of Step 2.21 The attainment of Step 1 targets within the cohorts is unlikely to reflect achievable attainment rates in a contemporary population since risk factor control and use of medication have improved since conception of these cohorts.<sup>22</sup> Therefore, estimates for non-attainment rates were derived from the more contemporary EUROASPIRE-IV survey. This survey, conducted ≥6 months after the index event, assessed attainment of all Step 1 targets simultaneously within a single population. To avoid underestimation of the effect of Step 1, the average attainment rates for the low-risk region were used as estimates for attainment rates that would be achievable in all risk regions (see Supplementary material online, Table S4 and Figure S1). In cases where Step 1 attainment rates within the cohorts were lower, a number of patients equal to that difference were modelled as having achieved that Step 1 target. A more thorough description of the incorporation of nonattainment is available in the supplementary material (see Supplementary methods). Intensification of SBP control and DPI from Step 1 targets to Step 2 targets (e.g. from <140-<130 mmHg) was limited to those patients who met their initial Step 1 treatment goal for these respective treatments. Consequently, in the scenario with non-attainment of Step 1, the Step 2 treatments could be applied to fewer patients. No information on Europe-wide SLGT2i initiation is available within EUROASPIRE surveys. Therefore, it was assumed that 25% of patients with DM and ASCVD would

receive SGLT2i in the scenario assuming non-attainment of Step 1. Assumptions for the exploratory analysis into additional lifestyle effects are available in the Supplementary methods. Due to limited data on dietary and activity habits, this exploratory analysis is restricted to the scenario assuming non-attainment only.

#### Impact of treatment on population level

The eligibility for each respective Step 2 treatment was calculated as percentage relative to the entire population. For each Step 2 treatment, the absolute risk reduction (ARR) was calculated (Supplementary material online, Table S3). The ARR was rescaled to account for simultaneous initiation of other treatments (see Supplementary methods). For those patients receiving a particular treatment, the ARR was translated into number needed to treat (NNT; 1/ARR) and the number of events saved per 1000 patients treated (1000/NNT) with that treatment. The impact of the stepwise approach was visualized using histograms and alluvial plots using the following risk categories: <10%, 10-20%, 20-30%, and ≥30%. Effects were analysed for the population as a whole and in mutually exclusive subgroups of patients. Mutually exclusive subgroups were made per CVD type. Patients with DM and ASCVD were considered a separate subgroup. In patients without DM, subgroups of patients with polyvascular disease [involving  $\geq 2$  of the following types of CVD: only CAD, only CeVD, or only PAD (including AAA; PAD/AAA), all without concomitant DM] were made.

Since Step 2 treatments are to be considered based on residual risk following Step 1, clinicians might maintain different threshold levels of residual risk at which to consider Step 2 treatments (e.g. a clinician might consider implementing Step 2 treatments only in patients with a residual risk of >15%). To demonstrate the impact of such a threshold, a dynamic web application was constructed. This application allows users to choose a threshold value of residual risk following Step 1 between 0% (treat all) and 30% (treat those with  $\geq$  30% CV event risk). Additionally, it allows for specifying which Step 2. treatments to include (e.g. leaving out low-dose colchicine from Step 2). Users can investigate the effect these choices on the main analyses displayed in this article. All treatment effects and risks in the main analyses otherwise specified. Estimates for risk are median with 25–75th percentile ranges, unless otherwise specified.

All analyses were performed using R Statistical Software Version 4.2.2. (Vienna, Austria).

## Results

# Study population and risk at current treatment

In total, 27 094 patients were included, 8730 of whom were part of the UCC-SMART cohort and 18364 were part of the REACH registry. Baseline blood pressure, use of blood pressure-lowering drugs, LDLc, and prevalence of polyvascular disease increased with higher levels of regional risk. Among those enrolled, CAD was the most common type of CVD, followed by CeVD and PAD/AAA. Peripheral artery disease was present in 20%, 24%, and 13% of patients, in the low-, moderate-, and (very) high-risk regions, respectively. Compared with patients without DM, patients with DM had on average higher SBP, triglyceride levels, longer duration of CVD, and higher body mass index (BMI) (see Supplementary material online, Tables S5 and S6). In the total study population, the overall 10-year risk of CV events with current treatment was 28% (18-47) and was 22% (15-32), 32% (22-46), and 60% (44-76) in the low-, moderate-, and (very) high-risk regions, respectively (Table 1). With current treatment, respectively, 7%, 1%, and 0.04% of patients in the low-, moderate-, and (very) high-risk regions were <10% risk of CV events (Figure 3).

# Step 2 in the presence of non-attainment in Step 1

In the scenario assuming non-attainment of Step 1 targets, the overall residual risk after Step 1 was 22% (14–36) and 17% (11–26), 24% (16–36), and 46% (32–63) for the low-, moderate-, and (very) high-risk

	Low risk (n = 15 610)	Moderate risk (n = 6266)	(Very) high risk (n = 5218)
Age (years)	64.5 (57–72)	68 (62–73)	63 (56–70)
Male, % (n)	75% (11739)	71% (4475)	67% (3499)
Systolic blood pressure (mmHg)	139 (20)	141 (19)	143 (20)
Systolic blood pressure at target, % (n)	61% (9543)	58% (3631)	54% (2809)
Use of antihypertensives, % (n)	82% (12830)	93% (5823)	95% (4958)
LDLc (mmol/L)	2.9 (1.0)	3.1 (1.1)	3.5 (1.1)
LDLc at target, % (n)	13% (2016)	8.6% (542)	5.2% (273)
Use of lipid-lowering therapy, % (n)	71% (11098)	74% (4629)	58% (3034)
Total cholesterol (mmol/L)	4.9 (1.1)	5.2 (1.1)	5.6 (1.2)
Triglycerides (mmol/L)	1.4 (1.0–2.0)	1.7 (1.2–2.3)	1.7 (1.2–2.4)
HDL cholesterol (mmol/L)	1.2 (0.4)	1.2 (0.4)	1.3 (0.4)
C-reactive protein (mg/dL)	2.2 (1.0-4.6)	2.4 (1.2–5.0)	2.4 (1.2–4.9)
eGFR (mL/min/1.73m <sup>2</sup> )	75 (18)	72 (19)	73 (20)
Years since first diagnosis	0.0 (0.0-6.0)	1.0 (0.0–10)	1.0 (0.0–9.0)
BMI (kg/m <sup>2</sup> )	27.3 (4.3)	28.0 (4.5)	28.1 (4.3)
Using antithrombotics, % (n)	76% (11930)	69% (4347)	76% (3959)
DM with ASCVD, % (n)	23% (3618)	38% (2389)	26% (1360)
Polyvascular disease no DM, % (n)	12% (1818)	13% (841)	18% (937)
CAD	10% (1597)	12% (760)	16% (845)
CeVD	6.1% (955)	8.0% (504)	10% (524)
PAD/AAA	7.9% (845)	12% (753)	7.0% (363)
Only CAD no DM, % (n)	39% (6082)	31% (1949)	39% (2008)
Only CeVD no DM, % (n)	16% (2466)	11% (701)	15% (790)
Only PAD/AAA no DM, % (n)	10% (1626)	6.2% (386)	2.4% (123)
Current smoking, % (n)	24% (3725)	18% (1118)	21% (1078)
10-year SMART2 risk	22% (15–32)	32% (22–46)	59 (44–76)

Table 1	Baseline characteristics across	the European risk regions
---------	---------------------------------	---------------------------

Summary statistics are mean and standard deviation (SD) or median (interquartile range) unless otherwise indicated. Targets are Step 1 targets. SMART2 risk is 10-year risk of recurrent cardiovascular disease. *n*, count; LDLc, low-density lipoprotein cholesterol; HDL, high-density lipoprotein; eGFR, estimated glomerular filtration; BMI, body mass index; CAD, coronary artery disease; CeVD, cerebrovascular disease; PAD, peripheral artery disease; AAA, abdominal aortic aneurysm; DM, type 2 diabetes mellitus; ASCVD, atherosclerotic cardiovascular disease.

regions, respectively. Following Step 2, residual risk decreased to 10% (6-18) overall and 8% (5-13), 10% (7-17), and 22% (14-34), for the respective risk regions. Of all Step 2 treatments, Step 2 LDLc reduction could be applied to the most patients regardless of residual risk threshold, followed by low-dose colchicine and DPI (Figure 1A). Apart from CVD subtype-specific therapies, applicability patterns for Step 2 treatments remained relatively comparable within CVD subtypes (see Supplementary material online, Figure S5). Depending on the threshold that was maintained, Step 2 could prevent up to 198, 223, and 245 additional events per 1000 patients treated, for the respective risk regions (Figure 1B). On a population level, the number of events prevented per 1000 patients for individual treatments was the highest for low-dose colchicine, followed by EPA, DPI, SBP < 130 mmHg, Step 2 LDLc reduction, and GLP-1 RA initiation (Figure 1B). Within CVD subtypes, the treatments with the highest number of events prevented per 1000 patients treated were colchicine when applicable or DPI or EPA in those where colchicine could not be considered (see Supplementary material online, Figure S5). Within CVD subtypes, the average residual risk following Step 1, as well as Step 2, was the highest for patients with polyvascular disease. Following Step 2, the residual risk was lowest for patients with CAD (Figure 2). The number of patients with a residual CV event risk <10% increased to 20%, 6.4%, and 0.5%, for the respective risk regions, following Step 1, and increased

to 63%, 48%, and 12%, respectively, following Step 2 (*Figure 3*). The effect of a threshold in residual risk after Step 1 for the implementation of Step 2 on all the analyses can be accessed at here.

### Exploratory analysis with extended lifestyle effects

In the scenario assuming non-attainment of Step 1 targets and extended lifestyle effects (increased physical activity and adherence to Mediterranean diet), the overall residual risk after Step 1 was 18% (11–31) and 14% (9–22), 20% (13–31), and 39% (26–56) for the low-, moderate-, and (very) high-risk regions, respectively. Following Step 2, residual risk decreased to 8.3% (5–15) overall and 6.6% (4–11), 8.5% (5–14), and 18% (11–29), for the respective risk regions. Other than the number of events prevented per 1000 patients treated being lower for each Step 2 treatment, compared to the main analysis, the pattern applicability and prevented events remained unchanged (see Supplementary material online, *Figure S7*). Within CVD subtypes, the effect of Step 1 was more pronounced, but Step 2 patterns remained unaffected (see Supplementary material online, *Figure S8*). The number of patients with a residual CV event risk <10% increased to 31%, 14%, and 1.8%, for the respective risk regions, following Step 1,



**Figure 1** Percentage treated (A) and events saved per 1000 patients treated (B) across residual risk thresholds across the European risk regions with non-attainment in Step 1. The three columns are for the three separate risk regions, with, on the first row, (A) the percentage of the total population in that region that receives a particular treatment and, on the second row, (B) the number of events saved per 1000 patients receiving the particular treatment. The x-axis represents the threshold in residual risk following Step 1, with, for example, an x-axis value at 10% in the low-risk region; all patients with a residual risk of 10% or higher are treated, meaning that 80% of the population would receive all Step 2 treatments for which they are eligible (A). Correspondingly when treating 1000 of these patients, 97 events might be prevented per 1000 patients treated (B). LDLc, low-density lipoprotein cholesterol; SBP, systolic blood pressure; GLP-1 RA, glucagon-like peptide 1 receptor agonists.

and increased to 71%, 56%, and 20%, respectively, following Step 2 (see Supplementary material online, *Figure S9*).

# Step 2 in the presence of perfect attainment of step 1

In the scenario assuming perfect attainment of Step 1 targets, the overall residual risk following Step 1 was 15% (9–24) and 11% (8–17), 16% (11–23), and 30% (21–43) for the low-, moderate-, and (very) high-risk regions, respectively. Following Step 2, overall residual risk decreased to 7% (4–12) and 5% (4–9), 7% (5–11), and 14% (9–22), for the respective risk regions. Compared with the scenario with non-attainment of Step 1, the Step 2 SBP reduction and DPI could be applied to more patients (see Supplementary material online, *Figure S2A*). Apart from CVD subtype-specific therapies, applicability patterns for Step 2 treatments remained relatively comparable within CVD subtypes (see Supplementary material online, *Figure S6*). Compared with the scenario with non-attainment of Step 1, the number of events prevented per 1000 patients treated was lower for Step 2 as a whole and each individual treatment when assuming perfect attainment of Step 1 (see Supplementary material online, Figure S2B). At a population level, Step 2 could prevent up to 181, 202, and 217 additional events per 1000 patients treated, for the low-, moderate-, and (very) high-risk regions, respectively, depending on the threshold that was maintained (see Supplementary material online, Figure S2B). The pattern with regard to the highest number of events prevented per 1000 patients treated within CVD subtypes was similar in both scenarios; however, compared with the scenario with non-attainment of Step 1, Step 2 treatments yielded fewer prevented events in the scenario with perfect attainment (see Supplementary material online, Figure S6). The pattern in reductions in residual risk within the CVD subtypes in the scenario assuming perfect attainment of Step 1 followed a similar pattern as to that in the scenario assuming imperfect attainment of Step 1 (see Supplementary material online, Figure S3). The number of patients in the low-, moderate-, and (very) high-risk regions <10% risk of CV events increased to 41%, 21%, and 3%, respectively, following Step 1, which further increased to 82%, 70%, and 30%, respectively, following Step 2 (see Supplementary material online, Figure S4).



**Figure 2** Average residual risk and risk reduction per treatment and CVD subtype across European risk regions with non-attainment in Step 1. The brackets refer to the combined effect of Steps 1 and 2, respectively. DM, diabetes mellitus; ASCVD, atherosclerotic cardiovascular disease; CAD, coronary artery disease; CeVD, cerebrovascular disease; PAD, peripheral artery disease; AAA, abdominal aortic aneurysm; LDLc, low-density lipoprotein cholesterol; SBP, systolic blood pressure; GLP-1 RA, glucagon-like peptide 1 receptor agonists.



**Figure 3** Changes in residual risk categories per treatment Step across the European risk regions with non-attainment in Step 1. The three separate plots are for the three separate risk regions. On the y-axis of each alluvial plot, the number of patients per risk region is seen. The stacked bars display the percentage of the total population with a risk smaller than the upper limit of each category (i.e. for 10–20%, all patients have a risk <20%). On the x-axis, the subsequent treatment steps can be seen (left to right): current treatment, Step 1, and Step 2. The stacked bar chart at each step on the x-axis refers to the residual risk following that Step (i.e. Step 2 is the distribution of risk following Step 2). Flows between bars illustrate movements of patients between categories following each treatment step.

## Discussion

The 10-year risk of recurrent CVD in patients with ASCVD varies across European risk regions and CVD subtypes. In the presence of non-attainment of Step 1, treatment targets up to 20% of patients are <10% risk for CV events; with Step 2, this could increase to 63%, and up to 245 CV events per 1000 patients treated could be prevented.

In recent decades, residual risk in patients with established ASCVD has captured significant attention, leading to concerted efforts aimed at its reduction. These efforts include investigations into stricter risk factor control, such as intensive SBP lowering, as in the SPRINT trial,

and intensive lipid-lowering, with, for example, PCSK9mabs, both of which were proven to be beneficial.<sup>23,24</sup> These two risk factors have long since been the cornerstones of CV risk reduction. Other paths to address residual risk have been investigated. For example, apart from CV risk reduction due to LDLc lowering, the effects of statins can also partly be ascribed to anti-inflammatory effects. This has prompted investigations into 'residual inflammatory risk', a concept that has subsequently been shown to provide useful treatment approaches in the LODOCO and CANTOS trials.<sup>25,26</sup> Furthermore, the COMPASS and PEGASUS trials demonstrated the effectiveness of intensive antithrombotic therapy and illustrated the concept of

'residual thrombotic risk'.<sup>27,28</sup> Additionally, the REDUCE-IT trial demonstrated the efficacy of EPA in patients with optimal statin therapy and it was hypothesized to be the result of both anti-inflammatory effects and lowering of residual triglyceride risk.<sup>29</sup> Novel drugs like GLP-1 RA possibly affect multiple of the aforementioned pathways by addressing obesity, which is part of the common soil for these risk factors.<sup>30</sup> The degree to which these multiple pathways contribute to the overall risk of a patient varies. While Step 1 sets targets for the population as a whole, Step 2 is aimed at personalization addressing these individual pathways. The new ESC guideline shaped this need for individualization into the stepwise approach and proposes considering the use of these novel treatments in selected individuals based on (predicted) residual risk.<sup>4</sup> The current study demonstrates the effects of this individualized approach at a population level, in real-world cohorts in patients with multiple CVD subtypes, and across multiple European risk regions. Additionally, the current study models the effect of Step 2 interventions in a combined and integrated approach, corresponding to clinical practice.

The high residual risk and the variability therein observed in the current study are in line with the few studies that estimate this risk in a secondary prevention population.<sup>1,31–34</sup> This high residual risk is partly due to delay in initiation of preventative treatments until the underlying disease has progressed to the point where clinical symptoms occur. Prior to the availability of novel treatments, guideline recommendations often only consisted of Step 1 targets and their intensification. The current analysis shows that even with perfect attainment of these targets, many patients remained at >10% risk of recurrent CV events. By implementation of Step 2 treatments, this residual CV event risk could be substantially reduced, both for patients already at Step 1 targets and those for whom these targets prove difficult to attain (e.g. smoking cessation). However, it is important to note that Step 1 interventions remain crucial in the management of ASCVD and should not be replaced by Step 2 treatment. The cohorts used in the analysis consist of patients receiving usual care; similar to Step 1, the effect of Step 1 here is smaller than in treatment-naïve patients. Notably, the reductions in risk from Step 2 treatment in patients with DM in combination with ASCVD are on average larger for the CVD specific treatments, like colchicine for CAD, then for DM-specific treatments like GLP-1 RA, indicating that these treatments should not be neglected in patients with DM and ASCVD.<sup>30</sup> Importantly, in patients eligible for DAPT, the risk reduction due to DPI would be comparable had they been prescribed DAPT, given the similar relative risk reductions.<sup>35</sup> However, Step 2 treatments should not be compared individually based on the current study since estimated effects could differ from real-world effects, for example, due to discontinuation due to side effects. Nonetheless, as illustrated in the current analysis, the combined initiation of Step 2 treatments is a potent strategy to reduce residual risk. Due to the advanced nature of atherosclerosis in these patients, CV event risk should be estimate on a regular basis to assess the need for treatment intensification.

The burden of CVD in Europe remains high, with a prevalence of 22 million, 25 million, and 7.8 million for, respectively, CAD, PAD, and CeVD in those countries included in this analysis alone.<sup>36</sup> Even with perfect attainment of Step 1 goals, ~4.3 million additional events might be prevented by treating all with Step 2 treatments for 10 years, assuming that these patients resemble those studied in this analysis. The effects of the stepwise approach on a population level have to be placed into the context of the providing healthcare system.<sup>37</sup> This is especially important given the differences in risk of recurrent CVD per geographical region which can partly be explained by differences in lifestyle, access to healthcare, and wealth. $^{38-40}$  In situations where resources limit the implementation of Step 2 interventions, policymakers might consider promoting access to relatively low-cost interventions such as colchicine or they might consider access to costly interventions to be restricted to special populations. Choices such as these are acknowledged in the guideline, which states that as treatment decisions

have an impact on healthcare costs and resources, countries or regions may decide on using higher or lower treatment thresholds.<sup>4</sup> The online application in the current study can help visualize the effect of limiting prescription of certain treatments to certain risk thresholds as well as omitting certain treatments from Step 2. In this way, insights from studies such as these might help guide decision-making. Regionally tailored treatment strategies may be necessary in different regions to optimize CV outcomes.

Strengths of the current study include the large cohorts with regional variation, the complementary scenarios, and the web application conforming to the individualized nature of Step 2. Limitations of this study should be considered. Among these are, firstly, the inability to incorporate lifestyle and weight loss explicitly, thus possibly mistakenly underrepresenting their importance. However, since no assumption was made on how LDLc and SBP reduction were achieved, their effects should reflect the effect of equivalent reductions attained by lifestyle intervention. Additionally, even in the exploratory analysis with the effect of highly optimistic estimates for lifestyle changes, the implementation of Step 2 results in substantial benefits. Secondly, due to the nature of evidence available, the effects of all Step 2 interventions had to be assumed to be independent from each other, possibly overestimating the effect of their combined implementation. Thirdly, the cohorts used for analyses could be considered somewhat historical, possibly resulting in differences in risk factor level control compared with patients today. However, since high-intensity statin therapy was available at the time, this is unlikely to be a substantial issue. Additionally, the scenario with perfect Step 1 attainment would be unaffected by such historical bias if it were present. While assumptions made on attainment might be seen as a limitation, the general principles, like the enhanced effectiveness of Step 2 treatments amidst Step 1 nonattainment, remain valid irrespective of true attainment. Not incorporating non-attainment would yield unrealistic estimates for the real-world efficacy of Step 2. Additionally, since non-attainment is likely to vary between populations and healthcare settings and systems, these assumptions have to be made regardless. Other limitations are the missingness in some predictor variables.

In conclusion, despite the current standards of care, residual CV risk in patients with ASCVD remains high across all European risk regions. Intensified Step 2 treatment as indicated in the 2021 ESC prevention guidelines results in marked further reduction of residual CV risk.

# Supplementary material

Supplementary material is available at European Journal of Preventive Cardiology.

## Acknowledgements

The authors gratefully acknowledge the contribution of the participants, the research nurses, Rutger van Petersen (data manager), Angela Vandersteen (study manager), and the members of the UCC-SMART study group and REACH registry investigators.

# **Author contribution**

J.A.N.D., F.L.J.V., and J.H. contributed to the design and data-analysis. J.H., F.L.J.V., J.A.N.D., and D.L.B. provided data for analysis. J.H., F.L.J.V., and J.A.N.D. drafted the manuscript. All authors critically revised the manuscript and contributed to design of analyses. All gave final approval and agreed to be accountable for all aspects of work ensuring integrity and accuracy.

## Funding

The UCC-SMART study was supported by a grant from the University Medical Centre Utrecht, The Netherlands. The REACH Registry was

supported by Sanofi-Aventis and Bristol-Myers Squibb and is endorsed by the World Heart Federation. The funding sources of the original studies had no involvement in the design and conduct of the current study; collection, management, analysis, and interpretation of the data; preparation, review, or approval of the manuscript; or decision to submit the manuscript for publication. The authors received no funding for the current study.

Conflict of interest: D.L.B.: advisory board: Angiowave, Bayer, Boehringer Ingelheim, CellProthera, Cereno Scientific, Elsevier Practice Update Cardiology, High Enroll, Janssen, Level Ex, McKinsey, Medscape Cardiology, Merck, MyoKardia, NirvaMed, Novo Nordisk, PhaseBio, PLx Pharma, and Stasys; board of directors: American Heart Association New York City, Angiowave (stock options), Bristol-Myers Squibb (stock), DRS.LINQ (stock options), and High Enroll (stock); consultant: Broadview Ventures, GlaxoSmithKline, Hims, SFJ, and Youngene; data monitoring committees: Acesion Pharma, Assistance Publique-Hôpitaux de Paris, Baim Institute for Clinical Research (formerly Harvard Clinical Research Institute, for the PORTICO trial, funded by St. Jude Medical, now Abbott), Boston Scientific (Chair, PEITHO trial), Cleveland Clinic, Contego Medical (Chair, PERFORMANCE 2), Duke Clinical Research Institute, Mayo Clinic, Mount Sinai School of Medicine (for the ENVISAGE trial, funded by Daiichi Sankyo; for the ABILITY-DM trial, funded by Concept Medical; for ALLAY-HF, funded by Alleviant Medical), Novartis, and Population Health Research Institute; Rutgers University (for the NIH-funded MINT trial); honoraria: American College of Cardiology (Senior Associate Editor, Clinical Trials and News, ACC.org; Chair, ACC Accreditation Oversight Committee), Arnold and Porter law firm (work related to Sanofi/Bristol-Myers Squibb clopidogrel litigation), Baim Institute for Clinical Research (formerly Harvard Clinical Research Institute; RE-DUAL PCI clinical trial steering committee funded by Boehringer Ingelheim; AEGIS-II executive committee funded by CSL Behring), Belvoir Publications (Editor in Chief, Harvard Heart Letter), Canadian Medical and Surgical Knowledge Translation Research Group (clinical trial steering committees), CSL Behring (AHA lecture), Cowen and Company, Duke Clinical Research Institute (clinical trial steering committees, including for the PRONOUNCE trial, funded by Ferring Pharmaceuticals), HMP Global (Editor in Chief, Journal of Invasive Cardiology), Journal of the American College of Cardiology (Guest Editor; Associate Editor), K2P (Co-Chair, interdisciplinary curriculum), Level Ex, Medtelligence/ReachMD (CME steering committees), MIH Life Sciences, Oakstone CME (Course Director, Comprehensive Review of Interventional Cardiology), Piper Sandler, Population Health Research Institute (for the COMPASS operations committee, publications committee, steering committee, and USA national co-leader, funded by Bayer), WebMD (CME steering committees), and Wiley (steering committee); other: Clinical Cardiology (Deputy Editor); Patent: sotagliflozin (named on a patent for sotagliflozin assigned to Brigham and Women's Hospital who assigned to Lexicon; neither I nor Brigham and Women's Hospital receive any income from this patent); research funding: Abbott, Acesion Pharma, Afimmune, Aker Biomarine, Alnylam, Amarin, Amgen, AstraZeneca, Bayer, Beren, Boehringer Ingelheim, Boston Scientific, Bristol-Myers Squibb, Cardax, CellProthera, Cereno Scientific, Chiesi, CinCor, Cleerly, CSL Behring, Eisai, Ethicon, Faraday Pharmaceuticals, Ferring Pharmaceuticals, Forest Laboratories, Fractyl, Garmin, HLS Therapeutics, Idorsia, Ironwood, Ischemix, Janssen, Javelin, Lexicon, Lilly, Medtronic, Merck, Moderna, MyoKardia, NirvaMed, Novartis, Novo Nordisk, Otsuka, Owkin, Pfizer, PhaseBio, PLx Pharma, Recardio, Regeneron, Reid Hoffman Foundation, Roche, Sanofi, Stasys, Synaptic, The Medicines Company, Youngene, and 89Bio; royalties: Elsevier (Editor, Braunwald's Heart Disease); site co-investigator: Abbott, Biotronik, Boston Scientific, CSI, Endotronix, St. Jude Medical (now Abbott), Philips, SpectraWAVE, Svelte, and Vascular Solutions; trustee: American College of Cardiology; and unfunded research: FlowCo. K.K.R. has received grants from Amgen, Sanofi, Regeneron, Daiichi Sankyo, and Ultragenyx; consulting fees from Novartis, Daiichi Sankyo, Kowa, Esperion, Novo Nordisk, MSD, Lilly,

Silence Therapeutics, AZ, NewAmsterdam Pharma, Bayer, Beren Therapeutics, CLEERLY, EMENDOBIO, SCRIBE, CRISPR, VAXXINITY, Amarin, Regeneron, Ultragenyx, Cargene, and Resverlogix; and speaking fees from Novartis, BI, AZ, Novo Nordisk, Viatris, Amarin, Biologix Pharma, Sanofi, Amgen, Esperion, Daiichi Sankyo, and Macleods Pharma, is president of the EAS, and hold stocks in NewAmsterdam Pharma and PEMI31. P.G.S. has received grants from Amarin and Sanofi; was involved in clinical trials (steering committee, CEC, DSMB): Amarin, Amgen, AstraZeneca, Bayer, BMS, Idorsia, Janssen, Novartis, and Sanofi; has received consulting/speaking fees from Amarin, Amgen, BMS, Novartis, and Novo Nordisk; and is a Senior Associate Editor at *Circulation*. The other authors have nothing to declare.

### Data availability

The data underlying this article will be shared on reasonable request to the corresponding author.

### References

- Kaasenbrood L, Boekholdt SM, Van Der Graaf Y, Ray KK, Peters RJG, Kastelein JJP, et al. Distribution of estimated 10-year risk of recurrent vascular events and residual risk in a secondary prevention population. *Circulation* 2016;**134**:1419–1429.
- Graham I, Atar D, Borch-Johnsen K, Boysen G, Burell G, Cifkova R, et al. European guidelines on cardiovascular disease prevention in clinical practice: executive summary. *Atherosclerosis* 2007;**194**:1–45.
- Smith SC, Allen J, Blair SN, Brass LM, Fonarow GC, Grundy SM, et al. AHA/ACC guidelines for secondary prevention for patients with coronary and other atherosclerotic vascular disease: 2006 update—endorsed by the National Heart, Lung, and Blood Institute. *Circulation* 2006;113:2363–2372.
- Visseren FLJ, Mach F, Smulders YM, Carballo D, Koskinas KC, Bäck M, et al. 2021 ESC guidelines on cardiovascular disease prevention in clinical practice. Eur Heart J 2021;42: 3227–3337.
- Ohman EM, Bhatt DL, Steg PG, Goto S, Hirsch AT, Liau C-S, et al. The REduction of Atherothrombosis for Continued Health (REACH) registry: an international, prospective, observational investigation in subjects at risk for atherothrombotic events-study design. Am Heart J 2006;151:786.e1–786.e10.
- Castelijns MC, Helmink MAG, Hageman SHJ, Asselbergs FW, de Borst GJ, Bots ML, et al. Cohort profile: the Utrecht Cardiovascular Cohort–Second Manifestations of Arterial Disease (UCC-SMART) study–an ongoing prospective cohort study of patients at high cardiovascular risk in the Netherlands. *BMJ Open* 2023;**13**:e066952.
- Hageman SHJ, McKay AJ, Ueda P, Gunn LH, Jernberg T, Hagström E, et al. Estimation of recurrent atherosclerotic cardiovascular event risk in patients with established cardiovascular disease: the updated SMART2 algorithm. Eur Heart J 2022;43:1715–1727.
- Mons U, Müezzinler A, Gellert C, Schottker B, Abnet CC, Bobak M, et al. Impact of smoking and smoking cessation on cardiovascular events and mortality among older adults: meta-analysis of individual participant data from prospective cohort studies of the CHANCES consortium. BMJ 2015;350:h1551.
- Ettehad D, Emdin CA, Kiran A, Anderson SG, Callender T, Emberson J, et al. Blood pressure lowering for prevention of cardiovascular disease and death: a systematic review and meta-analysis. *Lancet* 2016;**387**:957–967.
- Baigent C, Blackwell L, Emberson J, Holland LE, Reith C, Bhala N, et al. Efficacy and safety of more intensive lowering of LDL cholesterol: a meta-analysis of data from 170 000 participants in 26 randomised trials. *Lancet* 2010;**376**:1670–1681.
- Mihaylova B, Emberson J, Blackwell L, Keech A, Simes J, Barnes EH, et al. The effects of lowering LDL cholesterol with statin therapy in people at low risk of vascular disease: meta-analysis of individual data from 27 randomised trials. *Lancet* 2012;**380**:581–590.
- Collins R, Peto R, Hennekens C, Emberson J, Godwin J, Baigent C, et al. Aspirin in the primary and secondary prevention of vascular disease: collaborative meta-analysis of individual participant data from randomised trials. *Lancet* 2009;**373**:1849–1860.
- McGuire DK, Shih WJ, Cosentino F, Charbonnel B, Cherney DZI, Dagogo-Jack S, et al. Association of SGLT2 inhibitors with cardiovascular and kidney outcomes in patients with type 2 diabetes: a meta-analysis. JAMA Cardiol 2021;6:148–158.
- Sattar N, Lee MMY, Kristensen SL, Branch KRH, Del Prato S, Khurmi NS, et al. Cardiovascular, mortality, and kidney outcomes with GLP-1 receptor agonists in patients with type 2 diabetes: a systematic review and meta-analysis of randomised trials. *Lancet Diabetes Endocrinol* 2021;9:653–662.
- Arnold S V, Kosiborod M, Tang F, Zhao Z, Maddox TM, McCollam PL, et al. Patterns of statin initiation, intensification, and maximization among patients hospitalized with an acute myocardial infarction. *Circulation* 2014;**129**:1303–1309.

- Cannon CP, Khan I, Klimchak AC, Reynolds MR, Sanchez RJ, Sasiela WJ. Simulation of lipid-lowering therapy intensification in a population with atherosclerotic cardiovascular disease. JAMA Cardiol 2017;2:959–966.
- Ballantyne CM, Banach M, Mancini GBJ, Lepor NE, Hanselman JC, Zhao X, et al. Efficacy and safety of bempedoic acid added to ezetimibe in statin-intolerant patients with hypercholesterolemia: a randomized, placebo-controlled study. *Atherosclerosis* 2018; 277:195–203.
- Schmidt AF, Carter JP, Pearce LS, Wilkins JT, Overington JP, Hingorani AD, et al. PCSK9 monoclonal antibodies for the primary and secondary prevention of cardiovascular disease. *Cochrane Database Syst Rev* 2020;**2020**:1465–1858.
- Cannon CP, Blazing MA, Giugliano RP, McCagg A, White JA, Theroux P, et al. Ezetimibe added to statin therapy after acute coronary syndromes. N Engl J Med 2015;372: 2387–2397.
- 20. Nystoriak MA, Bhatnagar A. Cardiovascular effects and benefits of exercise. Front Cardiovasc Med 2018;5:135.
- Carbone S, Canada JM, Billingsley HE, Siddiqui MS, Elagizi A, Lavie CJ. Obesity paradox in cardiovascular disease: where do we stand? Vasc Health Risk Manag 2019;15:89–100.
- 22. Kotseva K, Wood D, De Bacquer D, De Backer G, Rydén L, Jennings C, et al. EUROASPIRE IV: a European Society of Cardiology survey on the lifestyle, risk factor and therapeutic management of coronary patients from 24 European countries. Eur J Prev Cardiol 2016;23:636–648.
- Berkelmans GFN, van der Graaf Y, Dorresteijn JAN, de Borst GJ, Cramer MJ, Kappelle LJ, et al. Decline in risk of recurrent cardiovascular events in the period 1996 to 2014 partly explained by better treatment of risk factors and less subclinical atherosclerosis. Int J Cardiol 2018;251:96–102.
- 24. A randomized trial of intensive versus standard blood-pressure control. N Engl J Med 2015; **373**:2103–2116.
- Steg PG, Szarek M, Bhatt DL, Bittner VA, Brégeault M-F, Dalby AJ, et al. Effect of alirocumab on mortality after acute coronary syndromes. *Circulation* 2019;**140**:103–112.
- Ridker PM, Everett BM, Thuren T, MacFadyen JG, Chang WH, Ballantyne C, et al. Antiinflammatory therapy with canakinumab for atherosclerotic disease. N Engl J Med 2017;377:1119–1131.
- Nidorf SM, Fiolet ATL, Mosterd A, Eikelboom JW, Schut A, Opstal TS, et al. Colchicine in patients with chronic coronary disease. N Engl J Med 2020;383:1838–1847.
- Bonaca MP, Bhatt DL, Cohen M, Steg PG, Storey RF, Jensen EC, et al. Long-term use of ticagrelor in patients with prior myocardial infarction. N Engl J Med 2015;372: 1791–1800.

- Steffel J, Eikelboom JW, Anand SS, Shestakovska O, Yusuf S, Fox KAA. The COMPASS trial: net clinical benefit of low-dose rivaroxaban plus aspirin as compared with aspirin in patients with chronic vascular disease. *Circulation* 2020;**142**:40–48.
- Bhatt DL, Miller M, Brinton EA, Jacobson TA, Steg PG, Ketchum SB, et al. REDUCE-IT USA: results from the 3146 patients randomized in the United States. *Circulation* 2020; 141:367–375.
- Marx N, Husain M, Lehrke M, Infodh V, Sattar N. GLP-1 receptor agonists for the reduction of atherosclerotic cardiovascular risk in patients with type 2 diabetes. *Circulation* 2022;**146**:1882–1894.
- 32. Quispe R, Bazo-Alvarez JC, Burroughs Peña MS, Poterico JA, Gilman RH, Checkley W, et al. Distribution of short-term and lifetime predicted risks of cardiovascular diseases in Peruvian adults. J Am Heart Assoc 2015;4:e002112.
- di Castelnuovo A, Costanzo S, Persichillo M, Olivieri M, de Curtis A, Zito F, et al. Distribution of short and lifetime risks for cardiovascular disease in Italians. Eur J Prev Cardiol 2012;19:723–730.
- 34. Marma AK, Berry JD, Ning H, Persell SD, Lloyd-Jones DM. Distribution of 10-year and lifetime predicted risks for cardiovascular disease in U.S. adults: findings from the National Health and Nutrition Examination Survey 2003 to 2006. *Circ Cardiovasc Qual Outcomes* 2010;**3**:8–14.
- Gynnild MN, Hageman SHJ, Dorresteijn JAN, Spigset O, Lydersen S, Wethal T, et al. Risk stratification in patients with ischemic stroke and residual cardiovascular risk with current secondary prevention. *Clin Epidemiol* 2021;**13**:813–823.
- Udell JA, Bonaca MP, Collet JP, Lincoff AM, Kereiakes DJ, Costa F, et al. Long-term dual antiplatelet therapy for secondary prevention of cardiovascular events in the subgroup of patients with previous myocardial infarction: a collaborative meta-analysis of randomized trials. Eur Heart J 2016;37:390–399.
- VizHub—GBD results. Accessed July 14, 2023. https://vizhub.healthdata.org/gbd-results/
- Joseph P, Leong D, McKee M, Anand SS, Schwalm J-D, Teo K, et al. Reducing the global burden of cardiovascular disease, part 1, Circ Res 2017;121:677–694.
- Li Y, Cao GY, Jing WZ, Liu J, Liu M. Global trends and regional differences in incidence and mortality of cardiovascular disease, 1990–2019: findings from 2019 global burden of disease study. *Eur J Prev Cardiol* 2022;**30**:276–286.
- Movsisyan NK, Vinciguerra M, Medina-Inojosa JR, Lopez-Jimenez F. Cardiovascular diseases in central and Eastern Europe: a call for more surveillance and evidence-based health promotion. *Ann Glob Health* 2020;86:21.